Author's response to reviews

Title: Patterns of research utilization on patient care units

Authors:

Carole A Estabrooks (carole.estabrooks@ualberta.ca)
Shannon Scott (shannon.scott@ualberta.ca)
Janet E Squires (janet.squires@ualberta.ca)
Bonnie Stevens (b.stevens@utoronto.ca)
Linda O'Brien-Pallas (l.obrien.pallas@utoronto.ca)
Judy Watt-Watson (j.watt.watson@utoronto.ca)
Joanne Profetto-McGrath (joanne.profeffot-mcgrath@ualberta.ca)
Kathy McGilton (McGilton.Kathy@TorontoRehab.on.ca)
Karen Golden-Biddle (kgbiddle@bu.edu)
Janice Lander (janice.lander@ualberta.ca)
Gail Donner (gdonner@utoronto.ca)
Geertje Boschma (boschma@nursing.ubc.ca)
Charles K Humphrey (humphrey@datalib.library.ualberta.ca)
Jack Williams (ji.williams@sympatico.ca)

Version: 4 Date: 27 May 2008

Author's response to reviews: see over
May 27th, 2008

Martin Eccles
Editor-in-Chief
Implementation Science

Dear Martin,

Re: MS: 1943758707155390
Patterns of research utilization on patient care units

As requested, we have made the suggested formatting changes.

I have uploaded the manuscript and the two additional files and two figures that accompany the manuscript.

Yours sincerely,

Carole A. Estabrooks
Professor & Canada Research Chair in Knowledge Translation